Ms. Monjur Ahmed  
Editor-in-Chief  
World Journal of Gastrointestinal Oncology

Dear Editor:

We wish to re-submit the manuscript titled “Association between intestinal neoplasms and celiac disease: a review”. The manuscript NO. is 64654.

We thank you and the reviewer for your thoughtful suggestions and insights. The manuscript has benefited from these insightful suggestions. I look forward to working with you and the reviewer to move this manuscript closer to publication in the World Journal of Gastrointestinal Oncology.

The manuscript has been rechecked and the necessary changes have been made in accordance with the reviewer’s suggestions. The responses to all comments have been prepared and attached in the manuscript.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,
Feng Gao  
Affiliation:  
Department of Gastroenterology, People's Hospital of Xinjiang Uygur Autonomous Region, Urumqi, Xinjiang, China.  
Address: NO. 91, Tianchi Road, Tianshan District, 830001, Urumqi, Xinjiang, China.  
Email: drxjgf@sina.com

Dear Reviewer #1:

Thank you for your comments concerning our manuscript. Those comments are valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Once again, thank you very much for your comments and suggestions.

The responds to your comments are as flowing:

1. For the diagnosis of celiac disease, at least one biopsy should be obtained from the duodenal bulb (ESPGHAN guideline). I recommend this statement should be added to "the Introduction part of the manuscript (second paragraph)".

Reply: Thank you for your suggestion. We have made correction according to your comment.